

Response to reviewers' letter

Dear Editor,

We greatly appreciate the efficient, professional and rapid processing of our paper by your team.

Be grateful for your elaborate guidance! The manuscript (NO.56287) was revised and some grammar and spelling errors had also been corrected. And I hope that you will find our study interesting and appropriate for your readers.

Response to Science Editors

(1) The “Author Contributions” section is missing. Please provide the author contributions.

Answer: Thanks to the careful guidance of the reviewers. We have added the “Author Contributions” section at the main text.

(2) The authors did not provide the approved grant application form(s). Please upload the approved grant application form(s) or funding agency copy of any approval document(s).

Answer: Thanks to the careful guidance of the reviewers. We have added funding agency copy of approval documents.

(3) The authors did not provide original pictures. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor.

Answer: Thanks to the careful guidance of the reviewers. We have provided the original figure documents using PowerPoint.

(4) PMID and DOI numbers are missing in the reference list. Please provide the PubMed numbers and DOI citation numbers to the reference list and list all authors of the references. Please revise throughout.

Answer: Thanks to the careful guidance of the reviewers. We have provided PMID and DOI numbers in the reference list.

(5) The “Article Highlights” section is missing. Please add the “Article Highlights” section at the end of the main text.

Answer: Thanks to the careful guidance of the reviewers. We have added the “Article Highlights” section at the main text.

Sincerely yours,

Qicai Liu,

Department of Reproductive Medicine Centre, 1st Affiliated Hospital, Fujian Medical University

20 Chazhong Road, Taijiang, Fuzhou 350005, China

Tel: 86-591-87981968

E-mail address: lqc673673673@163.com